Author's response to reviews

Title: Delayed endovascular treatment of descending aorta stent graft collapse in a patient treated for post-traumatic aortic rupture: a case report

Authors:

Giovanni Nano (giovanni.nano@libero.it)
Daniela Mazzacaro (danymazzak83@libero.it)
Giovanni Malacrida (gmalacrida@libero.it)
Maria T Occhiuto (mtocchiuto@yahoo.it)
Silvia Stegher (silvia.stegher@live.it)
Domenico G Tealdi (domenico.tealdi@unimi.it)

Version: 2 Date: 19 April 2011

Author's response to reviews:

In this new manuscript, we made all the corrections suggested by the reviewer. The new text was also correct in terms of grammar and spelling.